Author’s response to reviews

Title: High serum alkaline phosphatase levels, a study in 181 Thai adult hospitalized patients

Authors:

Viroj Wiwanitkit (wviroj@pioneer.netserv.chula.ac.th)

Version: 2 Date: 22 Aug 2001

PDF covering letter
'Author details' and 'Manuscript details' forms in web submission system:
******************************************************************************
********

1 'Author details' form - Author list matches that on the manuscript
2 'Author details' form - each author has a different email address
3 'Manuscript details' form - Title of manuscript correct matches that on
  the manuscript
4 'Manuscript details' form - Abstract matches that on the manuscript

Title page of manuscript:
******************************************************************************

5 Please check the form of your manuscript title. Titles should be as
concise and informative as possible. Ideally they should
include the type of intervention, the condition or type of participants or
subjects, and the methodology, as in the following examples:
A versus B in the treatment of C: a randomized controlled trial.
Or: X in the aetiology of Y: a case control study.
6 Authors' affiliations provided in the following format: Department,
Institute, City, Country
7 Each affiliation is linked to an author
8 All authors linked to their corresponding affiliation(s) using
superscript numerals
9 One corresponding author should be indicated

Manuscript sections:
******************************************************************************
10 We strongly suggest that your abstract is no longer than 250 words, as PubMed will truncate the record at 250 words
11 Abstracts should not cite references
12 Abstracts must be structured into Background, Methods, Results, Conclusions
13 Manuscript sections should include (in the following order): Abstract, Background, Methods, Results, Discussion, Conclusions, Acknowledgements, Competing Interests, References, Figure legends, Tables (if any), Description of Additional files (if any)
14 Please note: Figures must be uploaded as separate files (see below)

Competing interests:
*************************

15 We will require you to include a Competing Interests section in your manuscript; please consider the following points, write a declaration, and include it as a separate section of your manuscript located before the References.

In the context of medical research, a competing interest exists when your interpretation of data or presentation of information could possibly be influenced by non-scientific considerations. Such considerations may include financial gain or strongly held religious or political beliefs, although we do not insist that you declare anything
other than financial interests here.

If you are able to say 'no' to all these questions, your declaration should read 'none declared'.

Have you in the past five years received reimbursements, fees, funding, or salary from an organisation that may in any way gain or lose financially from the publication of this paper?

Do you hold any stocks or shares in an organisation that may in any way gain

Do you hold any stocks or shares in an organisation that may in any way gain or lose financially from the publication of this paper?

Do you have any other financial competing interests?

Are there any non-financial competing interests you would like to declare in relation to this paper?

References:
************

16 References must be cited in text using consecutive numbers in square brackets
17 Reference list should be provided in BioMed Central style - example
journal reference:

uainidinium thiocyanate-phenol-chloroform extraction.
Anal Biochem 1987, 162: 156-159

Figures:
*******

18 Each figure must be provided as a separate file using the 'Figure files'
business in the 'upload' page in web submission, not as part of the main
manuscript document
19 The image file should not include the legend (eg Figure 1....etc), which
should be part of the main manuscript after the References
20 Figures must be closely cropped, so only a small white border appears
around the actual image
21 Figures must be uploaded to the site in the order in which they are to
appear
22 Please use the 'preview' button on the 'upload' page in web submission to
examine the quality of your images, as these previews
examine the quality of your images, as these previews
will be identical to the final version we publish on the web - - if you do not check your preview we cannot guarantee to make changes to the figures at
a later date. If your figures are appearing badly in preview please contact
editorial@biomedcentral.com
Tables:

23 Tables less than one side of A4 can appear within the body of the article; if you wish them to do so they must be included at the end of your main manuscript document
24 Tables larger than this, or those that you do not wish to appear in the body of the article should be uploaded as additional files
25 Tables included as part of the main manuscript must be divided into cells
/ fields, the borders of which must be visible with black lines. Tables generated with tabbed text are not acceptable
generated with tabbed text are not acceptable

Additional Files:

26 May consist of larger tables or other files such as movies, pdf files, etc, that are not intended to appear within the body of the article
27 Additional files must have the appropriate three letter file extension
for the program you have used to generate them (eg. .xls for Excel;
.pdf for Acrobat files etc)
28 If you have uploaded additional files, you must include a separate section of manuscript listing: file name, file format, title of data, and short description of data
29 Additional files must be mentioned in the text, and referred to in the following way: 'see Additional file 1:[filename]....'
My submission to BioMed Central conforms to this checklist.

Name: Viroj Wiwanitkit